

Document de treball de l'IEB 2014/37

LOCAL INFRASTRUCTURES AND EXTERNALITIES: DOES THE SIZE MATTER?

Massimiliano Ferraresi, Umberto Galmarini, Leonzio Rizzo

Fiscal Federalism

Document de treball de l'IEB

LOCAL INFRASTRUCTURES AND EXTERNALITIES: DOES THE SIZE MATTER?

Massimiliano Ferraresi, Umberto Galmarini, Leonzio Rizzo

The **IEB** research program in **Fiscal Federalism** aims at promoting research in the public finance issues that arise in decentralized countries. Special emphasis is put on applied research and on work that tries to shed light on policy-design issues. Research that is particularly policy-relevant from a Spanish perspective is given special consideration. Disseminating research findings to a broader audience is also an aim of the program. The program enjoys the support from the **IEB-Foundation** and the **IEB-UB** Chair in Fiscal Federalism funded by Fundación ICO, Instituto de Estudios Fiscales and Institut d'Estudis Autonòmics.

The **Barcelona Institute of Economics (IEB)** is a research centre at the University of Barcelona (UB) which specializes in the field of applied economics. The **IEB** is a foundation funded by the following institutions: Applus, Abertis, Ajuntament de Barcelona, Diputació de Barcelona, Gas Natural, La Caixa and Universitat de Barcelona.

Postal Address: Institut d'Economia de Barcelona Facultat d'Economia i Empresa Universitat de Barcelona C/ John M. Keynes, 1-11 (08034) Barcelona, Spain Tel.: + 34 93 403 46 46 <u>ieb@ub.edu</u> <u>http://www.ieb.ub.edu</u>

The IEB working papers represent ongoing research that is circulated to encourage discussion and has not undergone a peer review process. Any opinions expressed here are those of the author(s) and not those of IEB.

LOCAL INFRASTRUCTURES AND EXTERNALITIES: DOES THE SIZE MATTER? *

Massimiliano Ferraresi, Umberto Galmarini, Leonzio Rizzo

ABSTRACT: We setup a model in which the residents of two neighboring municipalities can use the services provided by public infrastructures located in both jurisdictions. If services are either complements or substitutes in use, the municipalities strategically interact when investing in infrastructures; moreover, when they differ in population size, the small municipality reacts more to the expenditure of its neighbor than the big one. The theoretical predictions are then tested by estimating the determinants of the stock of public infrastructures of the municipalities belonging to the Autonomous Province of Trento, in Italy.

JEL Codes: D72, H61, H77

Keywords: Local public goods spillovers, spatial interactions, size of local jurisdictions

Massimiliano Ferraresi University of Ferrara Department of Economics and Management Via Voltapaletto 11, 44121 Ferrara, Italy E-mail: massimiliano.ferraresi@unife.it

Leonzio Rizzo University of Ferrara & IEB Department of Economics and Management Via Voltapaletto 11, 44121 Ferrara , Italy E-mail: <u>leonzio.rizzo@unife.it</u> Umberto Galmarini University of Insubria – Como & IEB Department of Law, Economics and Cultures Via Garibaldi 61-63, 22100 Como, Italy E-mail: <u>umberto.galmarini@uninsubria.it</u>

^{*} We wish to thank the Autonomous Province of Trento for kindly providing the data, and Piero Giarda, Ian Preston and seminar participants in Siegen, Barcelona and Lecce, at the 2014 Gerard-Varet Conference, Aix-en-Provence, the 2014 APET Conference, Seattle, the 2014 IIPF Conference, Lugano, for useful comments. Massimiliano Ferraresi gratefully acknowledges funding from the University of Ferrara; Umberto Galmarini and Leonzio Rizzo thankfully acknowledge financial support from the Spanish Ministry of Economy and Competitiveness (ECO2012-37873).

1 Introduction

Is the provision of public infrastructures by a local jurisdiction affected by that of its neighbors? And how is the effect (if any) related to the size of local jurisdictions in terms of population? A proper answer to these questions can give an important contribution to the discussion about the optimal boundaries of areas over which public infrastructures are provided. In fact, this is a hot topic in Europe, where some countries are rethinking the structure of their public sector (in terms of both the number and the types of government layers) by relying on two main theoretical arguments, namely the presence of scale economies and of positive spillovers in infrastructures provision, both pointing at inefficient levels of infrastructure provision by local jurisdictions that are too small in size.¹

The theoretical literature on fiscal externalities recognizes that there are various ways in which decisions taken in one jurisdiction may spill-over into other jurisdictions.² Fiscal policies of regional governments can directly affect the welfare of residents in neighboring jurisdictions, as for expenditures on public goods and services (e.g., environmental policies) whose benefits transcend borders. Public policies in one region can also indirectly affect residents elsewhere through their impact on local governments' budgets, giving rise to the so-called fiscal externalities (e.g., tax policies that induce tax base mobility across jurisdictions). Case et al. (1993) is the first systematic empirical work addressing these issues; using data on expenditures of continental US States over the period 1970-1985, they find that state government's per capita expenditure is positively and significantly affected by that of its neighbors'. Other important studies are Murdoch et al. (1993) and Solé-Ollé (2006), showing that public expenditure spillovers are stronger at low levels of government's layers than at high levels.

There is also a growing literature on fiscal externalities specifically related to the provision of local infrastructures. Cremer et al. (1997) modelling the provision of local infrastructures in a federation in which two communities strategically interact by comparing the per capita cost of providing infrastructures with the transport cost that their own citizens must bear to go and enjoy the infrastructures provided by the neighboring community. For given production and transport costs, the decision to provide

¹In the celebrated Decentralization Theorem by Oates (1972), the exploitation of scale economies and the internalization of spillovers account for the benefits of centralization, while uniform public goods provision in the presence of heterogenous preferences at the local level account for its costs.

 $^{^2 {\}rm For}$ a comprehensive analysis of the issue, see, e.g., Dahlby (1996).

infrastructures depends on the size of the community. Haughwout (2002) proposes a spatial equilibrium model by considering the role of infrastructures in determining the distribution of economic activity across regions. The empirical evidence suggests that central cities' land prices are positively related with public infrastructures provision; however, as the same author points out, the omission from the model of the costs and benefits of spillovers might be one of the main causes of the limited local infrastructure benefits found in the empirical analysis. Buettner et al. (2004), by using German data on public expenditure of Lander governments, find that the agglomeration level has no effect on the per capita expenditure on infrastructures; in particular, there is no cost disadvantage, both for highly urbanized and for sparsely populated regions. Also in this case, however, one might argue that the results are driven by the assumption of no spatial interaction between local infrastructures. In fact, if spatial autocorrelation turns out to be an important expenditure determinant, not accounting for it can yield biased and inconsistent estimates for many of the determinants of the expenditure equation (Case et al., 1993; Revelli, 2002).

In our work, to set the stage for the empirical analysis, we build up a simple theoretical model in which two neighboring local jurisdictions independently provide public infrastructures. If local infrastructures can be consumed by the citizens of both jurisdictions, the model shows that each local government increases (respectively, reduces) its expenditure on infrastructures in response to an increase in its neighbor's expenditure if local infrastructures are complements (respectively, substitutes) in use by citizens. Public infrastructures like roads, bridges, or dams, are examples of *complement infras*tructures, since they share the property that their benefits from use are higher if also the neighboring jurisdictions provide the same type of infrastructures on their territory. If two neighboring jurisdictions provide good roads, and if roads are not used only for local trips (i.e., confined within the boundaries of a given jurisdiction) but also for inter-jurisdictional trips, then the benefits from road usage are higher for the residents of both jurisdictions than in the case in which only one of them provides good roads. In this sense, local roads, like other types of infrastructures, can be complement in use. On the contrary, public facilities like theaters, libraries, or sport grounds, are examples of substitute infrastructures, since the citizens of a given jurisdiction can use either the facilities provided in their own jurisdiction or those provided in the neighboring jurisdictions, but never both at the same time.

The theoretical model also shows that, in per capita terms, the size of the reaction

of expenditure on infrastructures to changes in the expenditure by the neighboring jurisdiction is decreasing, in absolute value, in the size of the local jurisdiction. That is, in per capita terms a highly populated jurisdiction hardly reacts to changes in infrastructures of a scarcely populated neighbor, since any given change in the per capita expenditure of a small jurisdiction has a negligible per capita impact, in terms of public goods spillovers, on the residents of a large jurisdiction.

In the empirical analysis, we use a dataset containing financial and socioeconomic variables for the 223 municipalities belonging to the Italian Autonomous Province of Trento. After constructing a measure of the stock of infrastructures provided by municipalities, we estimate their determinants by explicitly introducing a spatial lag-error component. We find robust evidence that some types of public infrastructures are of the complement type, since in small municipalities their level is positively affected by the level of infrastructures provided by the neighboring communities. However, and in accordance with our theoretical predictions, the spatial interaction tends to vanish for large municipalities.

The work is structured as follows. Section 2 presents the theoretical model. Section 3 describes the data, Section 4 outlines the estimation strategy and discusses the results. Section 5 concludes.

2 The theoretical model

Consider a regional economy composed of two municipalities, labelled $i = 1, 2.^3$ Let N_i be the population resident in jurisdiction i, and y_i its per-capita endowment of income, exogenously given. Income is used to consume private and local public goods, the latter financed with a local income tax. We assume that individuals cannot change their place of residence, although they can move to consume the public good provided in the neighboring municipality.

Consider, without loss of generality, community 1. The utility function of the rep-

³The fact of limiting the analysis to the case of only two jurisdictions obviously implies that each one of them is 'the neighbor' of the other one. We adopt such a simplified setup for analytical convenience. A richer, but also more complex, specification is that of the 'circular region', a formalization akin to that used in spatial models of product differentiation, in which the local jurisdictions are located along a circle, so that each one of them has two neighbors, one at its left and one at its right of the regional territory (see Solé-Ollé, 2006, for an application of such a type of framework).

resentative individual resident in municipality 1 is:

$$u_1(z_1, z_2) = \left(\alpha_1 - \frac{z_1}{2}\right) z_1 + \theta \left[\left(\alpha_1 - \frac{z_2}{2}\right) z_2 + \phi z_1 z_2\right] + \left(\beta_1 - \frac{x_1}{2}\right) x_1, \quad (1)$$

where z_i denotes the effective service-level of public infrastructures provided by municipality *i* on its territory, and x_1 the consumption of private goods. The parameters $\alpha_i > 0$ and $\beta_i > 0$ are a measure of the intensity of preferences for the consumption of public and private goods, respectively.⁴

The utility function (1) also contains two parameters, $\theta \in [0,1]$ and $\phi \in [-1,1]$, which are key for the analysis, and that are assumed to be identical in the two jurisdictions. The parameter θ represents a classical positive spillover of local public goods provision; at one end, $\theta = 1$ implies full spillover; at the other end, $\theta = 0$ implies no spillovers.⁵ A first interpretation is that a share θ of the residents in jurisdiction 1 fully enjoy the public infrastructures located in jurisdiction 2. A second interpretation is that all residents in jurisdiction 1 enjoy at a θ % rate the public infrastructures located in jurisdiction 2.

The parameter ϕ measures instead the degree of complementarity (if positive) or substitutability (if negative) in the use of public infrastructures provided by the two jurisdictions.⁶ For instance, road services provided by the two municipalities are complement in usage if drivers (e.g., commuters, or shoppers) must cross the border in a typical journey: in this case, it is $\theta > 0$, $\phi > 0$. Two swimming pools, one located in each municipality, are instead likely to be substitutes in usage: in this case, it is $\theta > 0$, $\phi < 0$. Finally, it is also possible that services provided are neither complements nor substitutes; in this case, it is $\theta > 0$, $\phi = 0$. For instance, a swimming pool in one

⁴Heterogeneity between jurisdictions in terms of the preference parameters α_i and β_i can be due to geographical factors, demographic factors (e.g., the share of elderly in total population), characteristics of the local economy, and so on.

⁵In line with the prevalent literature, we assume that the spillover is automatically determined by the provision of local infrastructures. It is possible to extend our framework to the more realistic case in which the effective level of enjoyment depends on usage levels, endogenously chosen by individuals of the two jurisdictions.

⁶Most models analyzing local public goods spillovers assume that the total amount of public goods enjoyed by the residents of any given jurisdiction is equal to a weighted sum of the 'home' and the 'neighbors' public goods supplies, which means that the public goods provided by different jurisdictions are perfect substitutes (in our model, this case is obtained by setting $\phi = -1$). The more general functional form of the utility function given in Eq. (1) is widely used in oligopolistic models with product differentiation (see, e.g., Singh and Vives, 1984).

jurisdiction, and a public library in the other one, are likely to be independent in usage by the residents of both jurisdictions.

In any given jurisdiction i, the per-capita effective service-level of public infrastructures depends on total expenditure, E_i , and on the total number of users, \tilde{N}_i , according to the function:⁷

$$z_i = \frac{E_i}{\tilde{N}_i^r}, \qquad i = 1, 2, \tag{2}$$

where $r \in [0, 1]$ and

$$\tilde{N}_1 = N_1 + \theta N_2, \qquad \tilde{N}_2 = N_2 + \theta N_1.$$
 (3)

Consistently with the utility function defined in Eq. (1), Eq. (3) shows that public infrastructures provided in a given jurisdiction are fully enjoyed by all its residents and by a share θ of the residents in the neighboring jurisdiction or, alternatively, that they are fully enjoyed by all its residents while they are enjoyed at a reduced θ % rate by all non-residents. In Eq. (2), the parameter r, which is identical for the two jurisdictions, captures the degree of congestion in the use of public infrastructures. At one end, r = 0implies non rivalry in the use of public services; at the other end, r = 1 implies that public services are fully rival. Note that while the parameter r defines the degree of rivalness in consumption within the boundaries of the jurisdiction in which the public good is provided, the spillover parameter θ defines the degree of non-rivalness in spatial terms, so that $\theta = 0$ defines a purely local good whereas $\theta = 1$ defines a purely 'national' good.

Let t_i be the per capita local tax and g_i an unconditional grant, in per-capita terms, received by municipality *i* from an upper tier of government (e.g., the regional government). By substituting the local government budget constraint, $N_i(t_i + g_i) = E_i$, into the representative individual's budget constraint, $x_i = y_i - t_i$, we obtain the local economy resource constraint:

$$x_i = y_i + g_i - \frac{E_i}{N_i}.$$
(4)

2.1 Investment in public infrastructures

Local policy makers simultaneously and independently set their own expenditures on infrastructures with the aim of maximizing the welfare of the representative resident.

⁷Note that the specification given in Eq. (2), by which the stock variable z_1 is a function of the flow variable E_i , implicitly assumes full depretiation of the expenditure in infrastructures within the time period. That is, we consider a simple static model instead of a more complex dynamic framework.

Formally, and considering, without loss of generality, municipality 1, the policy maker chooses public expenditure E_1 to maximize Eq. (1), subject to Eqs. (2) and (4) for i = 1, taking as given the public expenditure E_2 of municipality 2.

The first order condition of the given problem is:

$$\frac{\partial u_1}{\partial E_1} = (\alpha_1 - z_1 + \theta \phi z_2) \frac{\partial z_1}{\partial E_1} + (\beta_1 - x_1) \frac{\partial x_1}{\partial E_1} = 0,$$

that can be written as:

$$\left(\alpha_1 - \frac{E_1}{\tilde{N}_1^r} + \theta \phi \frac{E_2}{\tilde{N}_2^r}\right) \frac{1}{\tilde{N}_1^r} - \left(\beta_1 - y_1 - g_1 + \frac{E_1}{N_1}\right) \frac{1}{N_1} = 0.$$
(5)

By solving Eq. (5) with respect to E_1 , we obtain the best response (or reaction) function:

$$\tilde{E}_1(E_2, N_1, N_2) = \left(\frac{\alpha_1}{\tilde{N}_1^r} - \frac{\beta_1 - y_1 - g_1}{N_1} + \frac{\theta\phi}{\tilde{N}_1^r \tilde{N}_2^r} E_2\right) \left/ \left(\frac{1}{N_1^2} + \frac{1}{\tilde{N}_1^{2r}}\right).$$
(6)

The second order sufficient condition for a maximum holds true, since:

$$\frac{\partial^2 u_1}{\partial E_1^2} = -\left(\frac{\partial z_1}{\partial E_1}\right)^2 - \left(\frac{\partial x_1}{\partial E_1}\right)^2 = -\left(\frac{1}{N_1^2} + \frac{1}{\tilde{N}_1^{2r}}\right) < 0.$$

A similar best response function, denoted by $\tilde{E}_2(E_1, N_2, N_1)$, can be obtained for municipality 2. By combining the two functions, one can solve for the Nash equilibrium in the public expenditure levels of the two municipalities.⁸ We characterize the factors that determine the sign and the size of the slope of the reaction function (6). In fact, the latter represents the key interaction effect for the expenditure decisions of local governments that we try to assess in our empirical analysis.

2.2 The slope of the reaction function

By linearity of the best response function (6), it is immediate to obtain its slope as:

$$\frac{\partial \tilde{E}_1}{\partial E_2} = \frac{\theta \phi}{\tilde{N}_1^r \tilde{N}_2^r} \left/ \left(\frac{1}{N_1^2} + \frac{1}{\tilde{N}_1^{2r}} \right) \right.$$
(7)

Provided that the benefits of public infrastructures spill-over across jurisdictions (i.e., $\theta > 0$), Eq. (7) shows that the sign of the slope of the reaction function is

⁸If a stable Nash equilibrium exists, then it is unique, since the reactions functions are linear in the expenditure levels. On the normative side, it is possible to show that in general the expenditure decisions emerging in the Nash equilibrium are not efficient, since local policy makers do not internalize the positive spillovers of public expenditure accruing to the residents in the other municipality.

determined by the sign of the parameter ϕ , expressing complementarity (when positive) or substitutability (when negative) of public infrastructures services in the neighboring jurisdictions. The following proposition summarizes the result.

Proposition 1 Assume that local infrastructures services spill-over across jurisdictions (i.e., $\theta > 0$). If the services provided by the two jurisdictions are complements in consumption by citizens (i.e., $\phi > 0$), then the best response function relating the optimal public expenditure levels of one jurisdiction to the public expenditure levels of the other jurisdiction is positively sloped. It is instead negatively sloped if the services provided by the two jurisdictions are substitutes in consumption (i.e., $\phi < 0$).

Quite obviously, if public services provided by the two jurisdictions are neither complements nor substitutes in use (i.e., $\phi = 0$), then public expenditure in each jurisdiction is independent of that in the other jurisdiction, even in the presence of positive spillovers.

2.3 The role of population size

In this section we explore how the population size affects the terms of the strategic interaction between the two local governments. Let $e_i = E_i/N_i$ denote per capita public expenditure in jurisdiction *i*. Expressed in per capita terms, the slope of the best response function is equal to:

$$\frac{\partial \tilde{e}_1}{\partial e_2} = \left(\frac{N_2}{N_1}\right) \frac{\partial \tilde{E}_1}{\partial E_2}.$$

Using Eqs. (7) and (3), the latter expression can be written as:

$$\frac{\partial \tilde{e}_1}{\partial e_2} = \left[\left(\frac{N_2}{N_1} \right) \frac{\theta \phi}{(N_1 + \theta N_2)^r (N_2 + \theta N_1)^r} \right] \left/ \left(\frac{1}{N_1^2} + \frac{1}{(N_1 + \theta N_2)^{2r}} \right).$$
(8)

We are interested in examining how the size of the slope shown in Eq. (8) is affected by changes in N_1 , for given N_2 , by focusing on some specific, but relevant, cases. In particular, assuming maximal spillovers (i.e., $\theta = 1$), we consider in turn the two 'polar' cases of services that are fully non-rival (i.e., r = 0) and fully rival (i.e., r = 1) in consumption.

If $\theta = 1$ and r = 0, then Eq. (8) reduces to:

$$\left. \frac{\partial \tilde{e}_1}{\partial e_2} \right|_{\theta=1,r=0} = \frac{\phi N_2}{\frac{1}{N_1} + N_1}.$$
(9)

Eq. (9) shows that, for given ϕ and N_2 , the slope of the best response function is, in absolute value, decreasing in $N_1 > 1$, since:

$$\frac{\partial^2 \tilde{e}_1}{\partial e_2 \partial N_1}\Big|_{\theta=1,r=0} = -\phi \gamma N_2, \quad \text{where } \gamma = \frac{N_1 - 1}{1 + N_1} > 0 \text{ for } N_1 > 1.$$

Substituting for $\theta = 1$ and r = 1 into Eq. (8), and then differentiating with respect to N_1 , we get:

$$\frac{\partial^2 \tilde{e}_1}{\partial e_2 \partial N_1} \bigg|_{\theta=1,r=1} = -\phi \eta N_2, \quad \text{where } \eta = \frac{N_1^2 + (N_1 + N_2)(N_1 - N_2)}{\left[N_1^2 + (N_1 + N_2)\right]^2} > 0 \text{ iff } N_1 > \frac{N_2}{\sqrt{2}}.$$

The latter expression shows that, for given ϕ and N_2 , the slope of the best response function is, in absolute value, decreasing in N_1 , provided that $N_1 > N_2/\sqrt{2}$, i.e., that jurisdiction 1 is not too small relative to the neighboring jurisdiction 2.

The results are summarized in the following proposition.

Proposition 2 Assume maximal spillovers $\theta = 1$. If r = 0, that is if services of local infrastructures are non-rival in use, then the size of the slope of the best response function of jurisdiction i is, in absolute value, decreasing in its population N_i , for given population N_j of the neighboring jurisdiction j. If r = 1, that is if services are rival in use, then the size of the slope of the best response function of jurisdiction i is, in absolute value, decreasing in N_j , for given N_j , provided that $N_i > N_j/\sqrt{2}$.

The intuition behind these results is simple. In per capita terms, a large jurisdiction has little incentives to react to changes in the infrastructures level by a small neighboring community, as any given change in the per capita expenditure of the latter brings about benefit spillovers to the residents in the large community that are, in per capita terms, very small.⁹

3 The empirical analysis

The model presented in the previous section is tested by using a dataset on the 223 municipalities belonging to the Italian Autonomous Province of Trento. Italy counts

⁹Although no analytical solutions are available for the general case in which $r \in (0, 1)$ and $\theta < 1$, by means of numerical simulations it is possible to show that the results of Proposition 2 are robust. In particular, for r sufficiently close to zero, the slope of the best response function is, in absolute value, decreasing in N_1 , for given N_2 . Instead, for r sufficiently close to 1, it is decreasing in N_1 provided that N_1 is above a given threshold \overline{N} , with $\overline{N} < N_2$. Details of the simulations are available from upon request.

four administrative government layers: the central authority and, at the local level, Regions, Provinces and Municipalities. While most Regions and Provinces are ruled by 'ordinary' statutes, some of them — the 'autonomous' Regions and Provinces — are ruled by 'special' statutes.¹⁰ Autonomy means wider competencies on public functions than those attributed to Ordinary Regions and Provinces, as well as the right to cash almost all tax revenues that originate at the local level. In particular, the Province of Trento cashes 90% of all revenues from central taxes that originate on its territory, while the remaining 10% is withheld by the central government. Thanks to its autonomy, the Province of Trento is a very interesting experimental framework, as fiscal federal theory can be tested within a simple institutional setting in which there are only two government layers: the Province in the role of the central authority and the Municipalities in the role of local authorities, with the latter financing their expenditure functions with own revenues and transfers from the Province.

At the municipal level, own revenues include a property tax and a range of userfees,¹¹ while provincial transfers are in part of the 'specific' type (i.e., targeted to specific expenditure functions) and in part of the 'general' type, with the latter allocated by means of formulas based on fiscal needs and fiscal capacities of the municipalities. On the expenditure side, budgetary data distinguish between 'recurrent' and 'capital' outlays. Our focus is on the latter type of expenditures, since they build up over time the stock of public infrastructures.

3.1 Data and variables

The main variable in the dataset is the yearly capital expenditure, in real terms,¹² for the 223 municipalities over the period 1990-2007, divided into 12 functions that reflect investments on different types of infrastructures. We also collected data on the capital transfers granted by the Province of Trento to its municipalities, since this source of

¹⁰Italy counts five Autonomous Regions (Sicily and Sardinia, which are insular territories, and Valle d'Aosta, Trentino Alto-Adige, and Friuli Venezia-Giulia, which are northern boundary territories) and two Autonomus Provinces (Trento and Bolzano, making up the Trentino Alto-Adige Region).

¹¹For the period covered by our study, the main local tax at the municipal level is ICI (Imposta Comunale sugli Immobili), which is based on the cadastral value of real estates and on the market value of building lots. Minor taxes include a surcharge on the personal income tax and a surcharge on the tax on electricity consumption. User charges include waste collection and fees for public services such as public transport, nursery schools, and so on.

¹²We used the 2007 base year deflator for gross fixed capital formation computed by the "Autorità per l'Energia" (www.autorita.energia.it).

revenues is an important determinant of investment outlays.¹³ The provincial capital transfers are in part of the specific type (i.e., earmarked to specific infrastructural projects in one of the 12 expenditure functions) and in part of the general type (usually formula-based, with reference to measures of fiscal needs and fiscal capacities).¹⁴

We build a measure of the municipal capital stock (i.e., the endowment of infrastructures) by applying the perpetual inventory method (see, e.g. Goldsmith, 1951; Meinen et al., 1998), according to which the capital stock at time t is assumed to be equal to the capital stock at time t-1, net of depreciation (if any), plus gross investment (capital expenditure) at time t. In our benchmark definition of the capital stock, we consider year 2001 as the initial capital stock, given that all the control variables are available from 2001 to 2007: the initial 2001 capital stock is computed by summing the yearly expenditure flows over the twelve-year period from 1990 to 2001. Assuming no depreciation, the 2002 capital stock is then obtained by adding to the 2001 stock the 2002 expenditure, and similarly for the following years from 2003 to 2007. Hence we end up with a seven-year series of municipal capital stock for the period 2001-2007. However, the capital stock does not show great variance between years, since infrastructural investments that typically take several years to be completed usually appear in the municipal budgets as uniform annual quota of expenditures. Furthermore, not all the controls (see below) are either available for each one of the years 2001-2007 or, when available, they do not show great variance between years. Therefore, in the empirical analysis of Section 4 we use a cross-sectional dataset instead of a panel, by using as dependent variable the average value of the capital stocks for the period 2001-2007.¹⁵

¹³There is also a well known literature on the effects of grants on public expenditure, usually finding that grants can stimulate government expenditures more than monetary transfers to individuals of the same amount (Gramlich, 1977). Hence, a quota of the federal money sticks to the public sector instead of being distributed to citizens (the so-called *flypaper effect*). Interestingly, Wyckoff (1991) finds that capital expenditures are particularly sensitive to grants.

 $^{^{14}}$ Data on general transfers cover the period 1991-2007 and account for about 60% of capital expenditures of all municipalities. Specific transfers cover only the period 2001-2007 since for the period 1991-2000 there was no distinction between the two categories of transfers in budgetary data. For the period 2001-2007, total transfers account for about 68% of capital expenditure, with specific transfers accounting for about 60% of the total.

¹⁵To compute the per capita value of the 2001-2007 stock, we divide it by the average population over the same period. To test the robustness of the results, we built several different measures of the capital stock, and found no significant changes (the details are available upon request). In particular, we computed the initial capital stock in year 1994 (obtaining a 14-year series) and in year 2006 (obtaining a 2-year series), assuming no depreciation. Then we also considered linear depreciation rates of 2%,

The per capita value of the average capital stocks for the period 2001-2007 is about 16,671 euros. Table 1 shows that almost 75% of the total stock is concentrated on three expenditure functions, namely Administration & Management, Roads & Transport and Planning & Environment. Moreover, these three functions are also those for which, in every year considered, all municipalities have a positive expenditure. For these reasons, our empirical analysis focuses on the determinants of four measures of infrastructural endowments: the total stock and the three above mentioned functions.

Expenditure Function	Per Capita Value (euro)	Percentage on the total	Number of zeros	% of observations with zero
	(1)	(2)	(3)	(4)
Administration & Management	3,625.57	21.75	0	0.00
Municipal police	7.39	0.04	210	94.17
Justice	236.09	1.42	25	11.21
Education	966.39	5.80	20	8.97
Culture	463.25	2.78	8	3.59
Sport	947.73	5.68	18	8.07
Tourism	226.63	1.36	85	38.12
Roads & Transport services	4,688.47	28.12	0	0.00
Planning & Environment	4,172.41	25.03	0	0.00
Social welfare	816.42	4.90	6	2.69
Economic development	468.83	2.81	63	28.25
In-house productive services	51.83	0.31	152	68.16
TOTAL	16,671.01	100.00		

Table 1: Summary statistics on the infrastructural stocks

Turning to the control variables, we build up a measure of the provincial capital grants (grants) using the same method outlined above for the capital stock. As for the other variables, the dataset includes denoised an end of the province data that can be relevant determinants of infrastructural stocks. (3) The average altitude level, 0.14*** 0.08** (0.08**) account for the fact that providing public services in the mountains requires 'greater', hence more costly, infrastructures than in plains. The number of residents (population, the 2001-2007 average value) can capture the presence of scale economies in infrastructures provision. The

^{3%, 4%} and 5%, which are in line with those used in similar studies estimating the stock of public infrastructures, such as those carried by the World Bank (Agénor et al., 2005; Arestoff and Hurlin, 2006) and the IMF (Kamps, 2006). For the Italian case, Marrocu and Paci (2008) use a 4% depreciation rate to build a measure of the capital stock series for the period 1996-2003 at the regional level.

shares (also in this case we refer to the 2001-2007 average value) of inhabitants older than 65 (*aged*) and of those aged 0-5 (*child*) can account for some specific infrastructural needs (e.g., infant schools, nursing homes for elderly). The per capita number of houses (*houses*), from 2001 Census, can capture the demand for public infrastructures from resident households, as well as those linked to tourism activities, since the variable includes also holiday properties. The per capita number of employees (from 2001 Census) in both the public and the private sector, (*total employees*), as well as the per capita number of firms (from 2001 Census) in the private sector (*local units*), can proxy the demand for public infrastructures from the productive sector. Finally, in order to capture a possible link between expenditure variation and population change we include the population growth rate (*growth*), defined as the percentange difference between the population average in 2001-2007 and that in 1991-1997.¹⁶

A peculiarity of the Province is the presence of 16 'communities', each one formed by several contiguous municipalities belonging to an homogenous geographic and economic area. By means of their community, the municipalities jointly provide some public services that benefit the whole area covered by the community, thus realizing some economies of scale and spillover internalization. Since the community of affiliation can bear some weight in the investment decisions of a municipality, we include as a control a dummy variable for each community (*communities dummy*).

In order to control for the outliers, we compute the interquartile range (IQR) for all the dependent variables, picking up those observations (outliers) passing over the left or right boundary and defining accordingly a dummy variable (*outliers dummy*).¹⁷ Finally, we define a dummy variable (*metropolitan dummy*), equal to one for the two most populated cities in the Province, which are Trento (about 110,000 inhabitants) and Rovereto (about 35,000 inhabitants). These are by far the biggest cities, since the other 221 municipalities have an average population of about 1,600 inhabitants.

Summary statistics, data description and data sources of all the variables used in the analysis are reported in Appendix A, Tables A1 and A2, while Table A3 provides the list of municipalities outliers.

 $^{^{16}\}mathrm{The}$ years 1991 and 2001 are the census years and 2007 is the last year of the dataset.

¹⁷We used the IQR stata command, which allows for the detection of both mild and severe ouliers.

4 Econometric specification

Estimation of the standard empirical model of public expenditure through a linear specification might not take into account expenditures and or economic shocks in neighboring municipalities which can be correlated with exogenous controls and so lead to biased and inconsistent estimates of the parameters (Case et al., 1993; Revelli, 2002). Therefore, before deciding upon the econometric strategy, we need to know whether the data present spatial dependence among municipalities.

To do so we first need to define spatial variables. We build a matrix of neighbors to each municipality based upon their geographical location, which can be expressed through a (223×223) matrix, such that the element corresponding to row a and column b is 1 if the spatial units a and b are geographically neighbors, and zero otherwise. We then make a row standardization such that the elements of each row sum to one; note Expenditure Function also that, since all neighbors have the same weight, all elements of a row are identical. (1) (2) (3) (4) Vector conditioned watagefiether (223 \times 2232) smatrix by the (223 \times 1) vector conditioned expenditure levels Vields for each municipality a Simple average of its 210 is 10 is 10 is 225 (11.21) expenditure. 966.39 5.80 20 8.97 We compute the traditional measure of spatial soft as $\frac{3.59}{10}$ Moran's spa- $\frac{947.73}{947.73}$ states and $\frac{3.278}{10}$ dependence that is the $\frac{3.59}{8.07}$ Moran's spager (2000) and (2

We compute the traditional measure of spatial dependence that is the 5.59 Moran's spatial sport sport compute the traditional measure of spatial dependence that is the 5.69 Moran's spatial statistics (Cliff and Ord, 1981; Anselin, 1988) for the per capita *Botal Infrastructures* and for the three selected sub-functions, $A_{25,03}^{28,12}$ inistration & Management, Roads & Transport value Planning & Environment 4.172.41 As Table 2 shows, all our variables of interest exibit a spatial pattern of postive autocorrelation $\frac{28,25}{1.18,000}$ for the per capita $\frac{28,25}{28,000}$ in *Total Infrastructures* and *Roads test* and *Roads test* for an approximate of the selected sub-function of the selected sub-functions of the selected sub-functions in *Total Infrastructure* and *Roads test* for an approximate of the selected sub-function of the selected selected sub-function of the selected selec

	Total	Administration & Management	Roads & Transport	Planning & Environment
	(1)	(2)	(3)	(4)
Moran I statistic	0.14***	0.08**	0.18***	0.06*
Notes: The spatial matrix	used to compute	he Moran test is a binary, contiguity	-based one, according to	o which two municipalities are

Table 2: Moran spatial statistic

Notes: The spatial matrix used to compute the Moran test is a binary, contiguity-based one, according to which two municipalities a neighbors if they share a border, and is row-standardized. *** significant at 1%; ** significant at 5%; *significant at 10%.

However, the result of the Moran test is unable to descriminate properly between spatial-lag and spatial-error dependence.¹⁸ Hence, in order to obtain a more precise

¹⁸There are two primary types of spatial dependence. The *spatial error dependence* occurs when the error terms across different spatial units are correlated. In this case the OLS assumption of uncorrelated error terms is violated and hence the estimates are biased. Spatial error is due to omitted (spatially

indication of which is the most likely source of spatial dependence, we perform the two robust Lagrange multiplier (LM) tests proposed by Anselin et al. (1996),¹⁹ which are based on the OLS residuals of a non-spatial regression model, using all the control variables described in Section 3.1. Table A4 shows that the robust LM tests indicate the presence of spatial lag dependence for *Total Infrastructures* (the robust LM test value is 3.53 and statistically significant at 10%) while for the *Roads & Transport* expenditure function the spatial pattern appears to be driven by both spatial lag and spatial error, even though the test statistic for the former (the robust LM test statistic is 9.50 and statistically significant at 1%) is larger than that for the latter (the robust LM test statistic is 6.05 and statistically significant at 5%). On the other hand, the robust LM test for the *Planning & Environment* expenditure function suggests the presence of spatial dependence in the error term (the test is equal to 5.44 and statistically significant at 5%), while for *Administration & Management* the robust LM test does not indicate the presence of neither spatial lag nor spatial error dependence.

For each one of the four infrastructural measures, we now proceed to estimate the slope of the reaction function characterized in Proposition 1, and then test for the results in Proposition 2, by estimating whether the size of the slope of the reaction function depends on population size.

4.1 Strategic interaction evidence

Using the per-capita neighbors' average expenditure, we first estimate the OLS coefficients (Table 3, columns 1, 3, 5 and 7). Moreover, since the spatial tests shown above suggest the presence of different patterns of spatial dependence for the four infrastructural measures we focus on, we perform a three-step procedure developed by Kelejian and Prucha (1998) to estimate a spatial autoregressive model with autoregressive disturbance (Table 3, columns 2, 4, 6 and 8) taking into account both the source of spatial dependence (spatial-lag and spatial-error) by using as instruments the average of all

correlated) covariates that, if not attended, would bias the estimate. Spatial error models sort out of the problem by estimating the coefficient of the spatial error. The *spatial lag dependence* implies that the dependent variable y in jurisdiction i is affected by independent variables of jurisdiction i and jand hence the dependent variable of j also affects it, and vice-versa. The assumption of uncorrelated error terms and independent observations is violated and therefore the regression estimates are biased. The solution to this puzzle can be that of instrumenting the endogenous spatial lag (i.e., the dependent variable of j entering in the estimate of the dependent variable of i).

¹⁹Both LM-statistics are Chi-Square distibuted with one degree of freedom.

neighbor's exogenous variables and correcting for heteroskedasticity of unknown form (GS2SLS Robust; see Drukker et al., 2010, 2011, for details).

The estimated spatial lag coefficient, λ , for Total Infrastructures is 0.17 and 1% significant while the spatial error coefficient, ρ , is not significant (Table 3, col. 2). The positive coefficient suggests that municipalities tend to increase their own infrastructure spending as a response to the rising expenditure of their neighboring municipalities, thus confirming the hypothesis of horizontal expenditure spillovers with complementarity in use; that is, in terms of the model in Section 2, the estimated coefficient λ is consistent with the existence of positive spillovers ($\theta > 0$) and complementarity in use ($\phi > 0$), which implies best responde functions that are positively sloped. As for the specific expenditure functions, for Road & Transport we find a 1% significant and positive coefficient of 0.25 for the spatial lag coefficient, λ , and a negative spatial error coefficient of 0.44 (5% singificant), indicating that both types of spatial dependence coexist in the model (Table 3, col. 6). For *Planning & Environment* the estimated spatial lag coefficient is not significant but in this case we find a negative and 5% significant spatial error coefficient of magnitude -0.47 (Table 3, col. 8). Finally, for the Administration & Management expenditure function we do not find any evidence of horizontal strategic interaction, since neither the spatial-lag nor the spatial-error coefficient are statistically different from zero (Table 3, col. 4). These results confirm those of the spatial auto-correlation tests illustrated in Tables 2 and A4. The reason is that the Road & Transport function contains investments on infrastructures that are complement in use, like roads and bridges, since the benefits from use of the infrastructure in one jurisdiction are higher if also the neighboring jurisdictions provide the same type of infrastructures on their territory. In the *Planning* \mathcal{B} *Environment* function, where the expenditure is related to infrastructures like dams, the spatial link in expenditure does not hold. However, there is a spatial link in the residuals of the estimate, which can be explained by some missing spatial variable reflected by the significance in the spatial error coefficient. Finally, in the Adminstration & Management function, which contains expenditures for buildings and facilities necessary to provide purely local administrative services, we do not find any spatial link, since there are no spillovers in use ($\theta = 0$, in terms of the model in Section 2).

	To	otal	Administration	& Management	Roads &	Transport	Planning &	Environment
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
λ	0.10*	0.17***	-0.17*	-0.03	0.14*	0.25***	-0.03	0.19
	(0.06)	(0.05)	(0.10)	(0.09)	(0.08)	(0.07)	(0.12)	(0.12)
ρ		-0.19		-0.17		-0.44**		-0.47**
		(0.16)		(0.15)		(0.17)		(0.19)
Grants	0.87***	0.85***	0.37**	0.37**	0.82***	0.79***	0.72***	0.74***
	(0.09)	(0.09)	(0.18)	(0.16)	(0.13)	(0.12)	(0.14)	(0.13)
Altitude	4.31***	3.83***	1.33	1.18	1.61**	1.44**	0.75	0.48
	(1.47)	(1.29)	(0.90)	(0.79)	(0.79)	(0.65)	(1.01)	(0.87)
Population	-0.07*	-0.07**	-0.06**	-0.06**	-0.03**	-0.04**	-0.03*	-0.03**
	(0.04)	(0.04)	(0.03)	(0.03)	(0.02)	(0.02)	(0.02)	(0.01)
Houses	2,509.36***	2,520.73***	446.31	391.00	633.78*	553.40*	1,290.00**	1,129.69***
	(883.25)	(812.42)	(527.35)	(491.88)	(380.58)	(334.17)	(501.19)	(416.22)
Aged	1,969.37	1,113.78	3,885.67	2,863.65	-1,603.27	-3,598.87	7,444.27	7,796.44
	(9,428.57)	(8,908.13)	(6,030.13)	(6,095.43)	(4,648.27)	(4,038.59)	(5,463.26)	(4,872.74)
Children	607.58	-3,717.58	-2,784.41	-5,084.36	-14,927.14	-16,478.47	11,248.30	5,203.81
	(30,717.96)	(28,556.94)	(18,234.70)	(17,034.96)	(20,203.51)	(18,083.26)	(16,124.65)	(14,392.43)
Population density	-0.83	-0.72	-0.46	-0.34	0.30	0.34	-0.31	-0.20
	(0.86)	(0.78)	(0.51)	(0.44)	(0.37)	(0.31)	(0.48)	(0.40)
Population growth	4,595.67	4,509.20	-2,203.88	-2,162.01	-1,762.93	-2,020.82	-416.09	-150.50
	(4,267.68)	(3,894.62)	(3,181.63)	(2,927.71)	(2,367.13)	(2,218.06)	(3,124.13)	(2,585.39)
Total employees	4,176.50	4,136.67	1,997.99	1,998.71	1,014.33	903.37	91.68	666.79
	(3,153.76)	(2,872.37)	(2,095.22)	(1,916.08)	(1,246.07)	(1,171.68)	(1,670.28)	(1,435.63)
Local unit	16,940.34	17,513.38	7,418.23	6,664.84	201.99	1,013.53	341.13	293.50
	(12,216.94)	(11,371.28)	(7,250.13)	(6,699.34)	(5,564.78)	(5,066.46)	(7,977.42)	(7,186.91)
Outliers dummy	10,167.77**	10,676.13***	8,286.30***	8,370.97***	7,490.02***	7,646.37***	8,541.15***	8,514.42***
	(4,115.77)	(3,944.14)	(1,171.37)	(1,141.15)	(1,011.15)	(906.81)	(1,671.18)	(1,443.07)
Metropolitan dummy	3,667.06	3,693.94	1,457.60	1,603.52	993.67	1,483.74	448.00	251.21
	(2,529.95)	(2,589.14)	(1,410.58)	(1,495.82)	(1,171.10)	(1,464.50)	(1,070.78)	(1,059.25)
Constant	-3,169.04	-3,278.43	-757.66	-598.66	1,138.46	1,408.10	-1,229.46	-1,821.05
	(2,911.58)	(2,688.47)	(1,975.98)	(1,909.38)	(1,689.41)	(1,466.95)	(1,711.60)	(1,469.92)
Communities dummy	YES	YES	YES	YES	YES	YES	YES	YES
Observations	223	223	223	223	223	223	223	223
R-squared	0.91		0.72		0.86		0.82	

Table	3:	Spatial	model	estimation	results
-------	----	---------	-------	------------	---------

Notes: Columns (1), (3), (5) and (7) display OLS robust estimator results by using, respectively, total per-capita expenditure, administration and management per-capita expenditure, road and traffic per-capita expenditure and planning and environment per-capita expenditure as dependent variables. Columns (2), (4), (6) and (8) show the spatial lag-error model estimation results by using GS2SLS (Generalized method of moment and instrumental variables) robust estimator correcting for heteroskedasticity issues of unknown form. The spatial weight matrix (W) used is of the type: contiguity-based and it is row-standardized. Robust standard errors are shown in parentheses. *** significant at 1%; ** significant at 5%; *significant at 10%.

4.2 The impact of the population size

In this section we extend our empirical analysis by interacting the average of neighbor's per capita expenditure with the population.²⁰ We first estimate a spatial model in which we account for the interaction term by OLS (Table 4, columns 1, 4, 7 and 10).²¹ We then use the GS2SLS estimator (Table 4, columns 2, 5, 8 and 11) where we account only for the endogeneity issue of the spatial parameter, but we do not instrument its interaction with the population. In order to check whether the results obtained from GS2SLS regression are robust to possible endogenity bias due to the interaction term we also include, as additional instruments, the product of the neighbor's exogenous variables with population (columns 3, 6, 9 and 12, Table 4).

The spatial interaction parameter, λ , is significantly different from zero for *To*tal Infrastructures in all three specifications: OLS (col. 1), GS2SLS (col. 2) where only neighbor's infrastructure are instrumented with all neighbors exogenous variables, and GS2SLS (col. 3) where neighbor's infrastructure are instrumented with all neighbors exogenous variables and the product of neighbors'infrastructure with population is instrumented with the product of neighbor's exogenous variables with population, confirming that municipalities' infrastructures are positively affected by those of the neighbors. Also for the Roads & Transports expenditure function λ remains significantly different from zero in all three specifications. We also find evidence of horizontal spending spillovers for Planning & Environment, since λ turns out to be significantly different from zero at 10% in the specification when we instrument only for neighbor's infrastructure (col. 11) and in the specification when both neighbor's infrastructure and the product of neighbors'infrastructure with population are instrumented (col. 12).

In columns 1, 2 and 3 of Table 4 we find evidence that the externality effect on *Total Infrastructures* is driven by population (in fact the intercation term *neighbors spending*population* is equal to -0.39, 5% significant in the specification in column 1; -0.45, 1% significant in the specification in column 2; -0.39, 5% significant in the specification in column 3), and then, inspecting more in detail we observe that this result is entirely due to the *Roads & Transport* expenditure function (columns)

²⁰To test the robustness of our results, we have also taken into account the relative size of a municipality with respect to its neighbors by interacting the average of neighbor's per capita expenditure with the ratio between population and the average of neighbor's population, finding no changes in the results. Results are available upon request.

²¹The results of the non-spatial regression model are reported in Table A5.

7, 8 and 9), for which the interaction term is negative (-1.17, 1%) singificant in the specification in column 7; -1.23, 1% significant in the specification in column 8 and -1.00, 1% significant in the specification in column 9). For *Planning & Environment* the interaction term is negative but, however, is statistically not different from zero (col. 10, col. 11 and 12). Finally, for *Administration & Management* the spatial coefficients and the interaction turns out to be statistically not different from zero in all the specifications, apart in the OLS specification (col. 4) where the spatial parameter λ is negative (-0.20) and statistically different from zero at 10%.

As for the spatial error coefficient, ρ , it is not significantly different from zero for *Total Infrastructures* (in both specifications at column 2 and 3) while it is significant for both *Road & Transport* (-0.44, 1% significant in the specification in col. 8 and -0.46, 1% significant in the specification in col. 9) and *Planning & Environment* (-0.41, 5% significant in the specification in col. 11 and -0.45, 5% significant in the specification in col. 12).

Our results show that *Total Infrastructures* are positively determined by the neighbor's infrastructure and so, by using Proposition 1, we can say that infrastructures affecting the provision of neighbor's infrastructures are complements: this result is strongly determined by *Roads & Transport* infrastructures. Moreover, the population of the municipality plays an important role, for both *Total Infrastructures* and *Roads & Transport*, in determining the size of the slope of the reaction function (confirming the result in Proposition 2); in particular, as the population increases the slope of the reaction function decreases.

		Total		Adminis	tration & Man	agement	R	oads & Transpo	ht	Plann	ing & Environ	nent
	(1)	(2)	(3)	(4)	(5)	(9)	(1)	(8)	(6)	(10)	(11)	(12)
Y	0.14^{**}	0.21^{***}	0.21***	-0.21*	00.0	-0.08	0.25***	0.35***	0.34***	0.01	0.21*	0.23*
	(0.06)	(0.05)	(0.05)	(0.11)	(0.0)	(0.0)	(0.08)	(0.07)	(0.07)	(0.14)	(0.13)	(0.13)
P		-0.18	-0.19		-0.24	-0.18		-0.44***	-0.46***		-0.41**	-0.45**
		(0.16)	(0.17)		(0.15)	(0.14)		(0.16)	(0.17)		(0.18)	(0.19)
Neighbors spending * Population	-0.39**	-0.45***	-0.39**	0.25	0.19	0.27	-1.17***	-1.23***	-1.00***	-0.27	-0.21	-0.15
	(0.18)	(0.17)	(0.16)	(0.25)	(0.20)	(0.23)	(0.33)	(0.29)	(0.27)	(0.23)	(0.23)	(0.23)
Grants	0.86***	0.84^{***}	0.84^{***}	0.37**	0.37^{**}	0.37^{**}	0.78***	0.76***	0.76***	0.71***	0.73***	0.73***
	(0.09)	(0.00)	(0.08)	(0.18)	(0.16)	(0.16)	(0.13)	(0.12)	(0.12)	(0.15)	(0.13)	(0.13)
Altitude	4.65***	4.22***	4.10^{***}	1.15	0.99	1.01	1.99**	1.79***	1.68***	0.65	0.37	0.34
	(1.42)	(1.25)	(1.25)	(0.91)	(0.80)	(0.79)	(0.77)	(0.64)	(0.63)	(1.01)	(0.87)	(0.87)
Population	0.48*	0.57**	0.47 **	-0.18	-0.15	-0.18*	0.36^{***}	0.37***	0.30***	0.05	0.03	0.01
	(0.26)	(0.24)	(0.23)	(0.12)	(0.0)	(0.11)	(0.11)	(0.10)	(0.00)	(0.07)	(0.06)	(0.07)
Houses	2,382.34***	2,369.45***	2,384.41***	484.14	389.59	441.10	512.95	428.70	453.67	1,281.75**	1,152.44***	1,142.47***
	(869.47)	(796.88)	(795.87)	(521.15)	(484.35)	(483.57)	(380.20)	(332.94)	(332.02)	(504.77)	(427.44)	(423.56)
Aged	1,906.98	1,288.11	1,195.29	4,063.03	2,407.21	3,055.88	-2,017.58	-3,781.36	-3,755.80	7,549.66	7,668.45	7,592.31
	(9,407.78)	(8,879.32)	(8,900.37)	(6,016.28)	(6,212.11)	(6,035.80)	(4,582.24)	(3,997.87)	(3,967.90)	(5, 415.54)	(4, 881.79)	(4,864.78)
Children	-1,315.34	-5,454.08	-5,799.35	-2,758.28	-6,246.95	-4,949.05	-18,838.30	-20,766.69	-20,361.58	11,001.59	5,930.14	5,517.65
	(30,629.61)	(28, 469.76)	(28,452.73)	(18, 311.44)	(16,960.47)	(17,131.07)	(19,908.48)	(17,752.45)	(17, 807.80)	(16,199.31)	(14,502.75)	(14, 432.38)
Population Density	-1.05	-0.96	-0.92	-0.44	-0.27	-0.32	0.10	0.17	0.20	-0.35	-0.23	-0.21
	(0.86)	(0.78)	(0.78)	(0.49)	(0.42)	(0.42)	(0.40)	(0.34)	(0.33)	(0.48)	(0.41)	(0.41)
Population growth	4,685.98	4,616.63	4,653.86	-2,210.55	-2,176.45	-2,174.15	-1,636.99	-1,770.11	-1,778.99	-445.50	-237.30	-209.57
	(4,269.57)	(3, 895.49)	(3,886.30)	(3, 191.27)	(2,900.26)	(2,915.19)	(2, 325.04)	(2, 181.09)	(2, 179.00)	(3, 119.88)	(2,631.02)	(2,603.18)
Total employees	3,869.87	3,879.20	3,899.29	2,174.53	2,082.59	2,161.76	1,012.47	1,043.16	1,022.35	115.09	580.74	621.57
	(3,137.68)	(2,837.23)	(2, 838.74)	(2, 163.75)	(1, 921.45)	(1,963.50)	(1, 234.55)	(1, 166.29)	(1, 164.82)	(1,669.06)	(1,461.46)	(1,453.64)
Local unit	19,179.96	19,763.99*	19,498.96*	7,042.96	6,155.76	6,377.81	1,315.12	1,603.29	1,532.70	615.30	368.31	272.37
	(12,207.45)	(11, 319.90)	(11, 310. 91)	(7,296.97)	(6,677.07)	(6,717.98)	(5,582.83)	(5,093.50)	(5,057.19)	(7,992.98)	(7,221.57)	(7, 198.25)
Outliers dummy	$10,254.36^{**}$	10,755.43***	$10,779.10^{***}$	8,291.36***	8,384.63***	8,374.56***	7,415.54***	7,492.34***	7,510.43***	8,533.51***	8,543.25***	8,553.18***
	(4, 118.36)	(3, 931.83)	(3,937.91)	(1, 171.67)	(1, 145.34)	(1, 135.08)	(973.47)	(874.58)	(873.39)	(1, 682.66)	(1,468.69)	(1,457.61)
Metropolitan dummy	91.18	-236.31	340.84	2,517.13	2,374.30	2,671.16	-1,630.26	-1,027.21	-517.47	-199.97	-176.38	-35.82
	(3, 733.13)	(3,759.75)	(3, 599.99)	(1,908.58)	(1, 579.04)	(1,739.14)	(1, 456.69)	(1,751.09)	(1,680.15)	(932.33)	(1,007.91)	(1,068.37)
Constant	-3,563.93	-3,706.10	-3,658.38	-476.14	-255.08	-296.57	1,167.71	1,517.63	1,508.55	-1,179.54	-1,696.90	-1,783.81
	(2,936.28)	(2,714.56)	(2,711.99)	(2,046.49)	(1,960.47)	(1,984.59)	(1,669.05)	(1, 447.14)	(1,443.67)	(1,719.73)	(1,507.46)	(1,503.76)
Communities dummy	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES
Observations	223	223	223	223	223	223	223	223	223	223	223	223
R-squared	0.91			0.73			0.87			0.82		

5 Conclusions

In this study we investigated the types of interactions that can emerge (if any) among municipalities in providing infrastructures. First, we setup a theoretical model in which two jurisdictions provide their own infrastructures, assuming that the inhabitants of both jurisdictions can use them. If local infrastructures are complements in use, there is a positive interaction when jurisdictions set their own expenditures; the interaction is instead negative if infrastructures are substitute in use by citizens. The model also predicts that an increase in population decreases the size of the reaction function slope. We then tested these results by using data on municipalities of the Italian Province of Trento, finding that total infrastructures of a jurisdiction are positively linked to neighbor's total infrastructures. This result holds also for some specific types of infrastructures, namely Roads & Transport and Planning & Environment, for which the municipalities show a complementarity relationship with their neighbors as it regards decisions on infrastructure provisions. Also the theoretical prediction about the impact of population on the stratetic response has been confirmed for the same type of infrastructural measures, since the size of the slope of the reaction function decreases in magnitude as population increases.

References

- Agénor, P.R., Nabli, M.K., Yousef T.M. (2005). Public Infrastructure and Private Investment in the Middle East and North Africa. World Bank Policy Research Working Paper, No.3661.
- Anselin, L. (1988). Spatial econometrics: Methods and models. Boston: Kluwer Academic Publisher.
- Anselin, L., Bera, K., Florax, R.J.G.M., and Yoon, M. (1996). Simple diagnostic tests for spatial dependence. *Regional Science and Urban Economics*, 26, 77–104.
- Anselin, L., and Florax, R.J.G.M. (1995). Small sample properties of tests for spatial dependence in regression models: Some further results. In L. Anselin and R.J.G.M. Florax, New directions in spatial econometrics (pp. 21–74). Berlin-Heidelberg: Springer.
- Arestoff, F., anf Hurlin, C. (2006), Estimates of Government Net Capital Stocks for 26 Developing Countries, 1970-2002. World Bank Policy Research Working Paper, No.3858.
- Buettner, T., Schwager, R., and Stegarescu, D. (2004). Agglomeration, population size, and the cost of providing public services: An empirical analysis for German States. *Public Finance and Management*, 4, 496–520.
- Case, A.C., and Rosen, H.S. (1993). Budget spillovers and fiscal policy interdependence: Evidence from the States. *Journal of Public Economics*, 52(3), 285–307.
- Cliff, A., and Ord, J. (1981). Spatial processes. Models and applications. London: Pion.
- Cremer, H., Marchand, M., and Pestieau, P. (1997). Investment in local public services: Nash equilibrium and social optimum. *Journal of Public Economics*, 65(1), 23–35.
- Dahlby, B. (1996). Fiscal externalities and the design of intergovernmental grants. International Tax and Public Finance, 3(3), 397–412.
- Drukker, D., Egger, P., and Prucha, I.R. (2010). On two-step estimation of a spatial autoregressive model with autoregressive disturbances and endogenous regressors. *Technical Report*, Department of Economics, University of Maryland.
- Drukker, D., Prucha, I.R., and Raciborski, R. (2011). A command for estimating spatial-autoregressive models with spatial-autoregressive disturbances and additional endogenous variables. *Technical report, Stata.*
- Goldsmith, R.W., (1951). A Perpetual Inventory of National Wealth, in Studies in Income and Wealth, M.R. Gainsburgh (eds), 14, 5-74. Princeton: Princeton University Press.
- Gramlich, E.M. (1977). Intergovernmental grants: a review of the empirical literature. In W.E. Oates (Ed.), *The political economy of fiscal federalism*, Lexington, MA (pp. 219–239).
- Haughwout, A.F. (2002). Public infrastructure investments, productivity and welfare in fixed geographic areas. *Journal of Public Economics*, 83(3), 405–428.
- Kamps, C. (2006). New Estimates of Government Net Capital Stocks for 22 OECD Countries, 1960-2001. IMF Staff Papers, 53(1), 120–150.
- Kelejian, H., and Prucha, I.R. (1998). A generalized spatial two-stage least squares procedure for estimating a spatial autoregressive model with autoregressive disturbances. The Journal of Real Estate Finance and Economics, 17, 99–121.

- Marrocu, E., and Paci, R. (2008). The effects of public capital on the productivity of the Italian regions. *Applied Economics*, 42 (8), 989–1002.
- Meinen, G., Verbiest, P., and deWolf, P. (1998). Perpetual inventory method: Service lives, discard patterns and depreciation methods. *Technical report, Statistics Netherlands*.
- Murdoch, J., Rahmatian, M., and Thayer, M. (1993). A spatially autoregressive median voter model of recreation expenditures. *Public Finance Quarterly*, 21, 334–350.
- Oates, W. E. (1972). Fiscal federalism. New York: Harcourt Brace Jovanovich.
- OECD (1993). Methods Used by OECD Countries to Measure Stocks of Fixed Capital, National Accounts: Sources and Methods. No. 2, Paris.
- Revelli, F. (2002). Testing the taxmimicking versus expenditure spill-over hypotheses using English data. *Applied Economics*, 34(14), 1723–1731.
- Singh, N., and Vives, X. (1984) Price and quantity competition in a differentiated duopoly. The RAND Journal of Economics, 15(4), 546–554
- Solé-Ollé, A. (2006). Expenditure spillovers and fiscal interactions: Empirical evidence from local governments in Spain. Journal of Urban Economics, 59(1), 32–53.
- Wyckoff, P.G. (1991). The elusive flypaper effect. Journal of Urban Economics, 30, 310-328.

APPENDIX

Table A1: Data description and data sources

Variables	Description	Source
Total per capita expenditure	Per capita stock of expenditure on infrastructure over the period 1990-2007 expressed in 2007 base year values by using deflator for gross fixed capital formation.	Province of Trento (PAT)
Neighbouring total per capita expenditure	Neighbouring average value of Total per capita expenditure on Infrastructure.	Our computation on PAT data
Administration & Management per capita expenditure	Per capita stock on Administration and Management expenditure function over the period 1990-2007 expressed in 2007 base year values by using deflator for gross fixed capital formation.	РАТ
Neighbouring Administration & Management per capita expenditure	Neighbouring average value of Road and Traffic per capita expenditure.	Our computation on PAT data
Roads & Transport per capita expenditure	Per capita stock on Road and Traffic expenditure function over the period 1990-2007 expressed in 2007 base year values by using deflator for gross fixed capital formation.	РАТ
Neighbouring Roads & and Transport per capita expenditure	Neighbouring average value of Road and Traffic per capita expenditure.	Our computation on PAT data
Planning & Environment per capita expenditure	Per capita stock on Planning and Environment expenditure function over the period 1990-2007 expressed in 2007 base year values by using deflator for gross fixed capital formation.	РАТ
Neighbouring Planning & Environment per capita expenditure	Neighbouring average value of Road and Traffic per capita expenditure.	Our computation on PAT data
Neighbors spending* population	Neighbouring total per capita expenditure and neighbouring per capita expenditure related to Administration and Management, Road and Traffic, Planning and Environment functions *population*10 ⁻³	Our computation on PAT data
Grants	Stock of total per capita grants over the period 1990-2007 and stock of per capita grants related to Administration and Management, Road and Traffic, Planning and Environment functions.	Our computation on PAT data
Population	Average population over the period 2001-2007.	ISTAT
Children	Average population between 0-5 years old over the period 2001-2007 divided by population	ISTAT
Altitude	Height of municipality above the level of the sea	ISTAT
Aged	Average population over 65 years old over the period 2001-2007 divided by population.	ISTAT
Houses	Number of houses in 2001 divided by population.	ISTAT
Population Density	Population divided by area.	Our computation
Total Employees	Number of public and private employees in 2001 divided by population.	ISTAT
Local Unit	Number of local productive unit in 2001 divided by population.	ISTAT
Outliers Dummy	Outliers dummy=1 if the municipality is an outliers with respect to per capita expenditure. See Table A3.	Our computation
Metropolitan Dummy	Metropolitan dummy=1 if the municipality is either Trento or Rovereto.	Our computation
Population Growth	Average population over the period 2001-2007 divided by average population over the period 1991-1997 minus one.	Our computation

Table A2: Descriptive statistics.

Variables	Obs	Mean	Std. Dev.	Min	Max
Total per capita expenditure	223	16,671.01	9,987.06	4,377.21	77,057.88
Neighboring total per capita expenditure	223	15,920.16	5,047.97	6,487.54	28,237.72
Neighboring total per capita expenditure*population*10 ⁻³	223	3,093.01	10,417.03	148.06	147,749.70
Administration & Management per capita expenditure	223	3,625.57	3,395.87	513.22	24,825.35
Neighboring Administration & Management per capita expenditure	223	3,521.14	1,783.10	1,032.28	9,032.59
Neighboring Administration & Management per capita expenditure*population*10 ⁻³	223	734.35	3,201.84	15.11	46,962.10
Roads & Transport per capita expenditure	223	4,688.47	3,928.27	589.79	25,643.57
Neighboring Roads & Transport per capita expenditure	223	4,277.21	2,047.51	1,156.27	11,686.00
Neighboring Roads & Transport per capita expenditure*population*10 ⁻³	223	781.91	2,466.72	39.05	34,585.22
Planning & Environment per capita expenditure	223	4,172.41	3,836.61	577.56	39,446.68
Neighboring Planning & Environment per capita expenditure	223	3,903.58	1,485.85	1,211.78	8,986.70
Neighboring Planning & Environment per capita expenditure*population*10 ⁻³	223	739.80	2,159.21	29.84	28,659.04
Total per capita grants	223	10,072.23	7,925.70	2,352.13	70,388.19
Administration & Management per capita grants	223	1,729.18	1,238.74	23.93	10,450.14
Roads & Transport per capita grants	223	2,189.10	2,157.26	264.42	17,314.44
Planning & Environment per capita grants	223	2,229.67	2,582.99	90.81	31,624.64
Altitude	223	709.40	294.15	85.46	1,491.12
Population	223	2,224.06	7,865.26	108.29	109,334.90
Houses	223	0.77	0.41	0.36	3.18
Aged	223	0.19	0.03	0.12	0.32
Children	223	0.06	0.01	0.03	0.10
Population Density	223	105.27	264.90	3.72	3,699.05
Population Growth	223	0.07	0.07	-0.18	0.31
Total employees	223	0.26	0.16	0.06	0.98
Local Unit	223	0.08	0.04	0.03	0.37

Notes: The spatial matrix used to compute the neighboring variables is a binary contiguity-based one, according to which two municipalities are neighbors if they share a border, and is row-standardized

Table A3: Municipalities outliers

	Total		Administration & Mar	nagement	Roads & Transport		Planning & Environment	
Q1	9507.48		1645.68		2255.12		2047.08	
Q3	20498.20		4212.22		5691.41		5134.31	
IQR	10990.72		2566.54		3436.28		3087.24	
Q1 –1.5*IQR e Q3 + 1.5*IQR	-6978.60	36984.28	-2204.13	8062.03	-2899.30	10845.83	-2583.78	9765.17
Number of mild outliers	9		10		6		3	
Q1 $-3*IQR e Q3 + 3*IQR$	-23464.68	53470.36	-6053.94	11911.84	-8053.73	16000.26	-7214.64	14396.03
Number of severe outliers	2		7		7		6	
	ALBIANO	44965.76	ALBIANO	24825.35	AMBLAR	15036.89	CENTA SAN NICOLO'	15202.88
	BRESIMO	41930.06	BOCENAGO	11175.95	BRESIMO	19096.17	DAONE	11027.09
	BRIONE	37741.92	BRIONE	13511.83	CAGNO'	15326.98	LARDARO	18115.57
	CASTEL CONDINO	42865.77	CADERZONE	8814.79	CASTEL CONDINO	17584.93	LUSERNA	12485.57
	GARNIGA	43060.11	CASTEL CONDINO	9132.70	DARE'	10885.35	MASSIMENO	20114.78
	GRAUNO	40976.11	CASTELFONDO	9509.00	GRAUNO	24137.27	PALU' DEL FERSINA	18117.79
	MASSIMENO	44222.98	CINTE TESINO	12137.98	MASSIMENO	12692.81	SAGRON MIS	11081.02
	PALU' DEL FERSINA	66917.53	DON	14464.66	PALU' DEL FERSINA	25643.57	TERRES	16701.57
Municipalities	PIEVE TESINO	39948.09	GARNIGA	20772.40	PRASO	11988.20	VIGNOLA FALESINA	39446.68
	SAGRON MIS	44057.73	GIUSTINO	8306.35	PREZZO	17483.32		
	VIGNOLA FALESINA	77057.88	GRUMES	11120.82	SAGRON MIS	22423.47		
			LONA-LASES	12860.40	STREMBO	12627.79		
			MASSIMENO	10839.78	VIGNOLA FALESINA	19491.20		
			PIEVE TESINO	22963.56				
			RONZONE	9163.028				
			TIARNO DI SOTTO	10135.25				
			VIGNOLA FALESINA	8643.582				

Table A4: Non spatial model – OLS regression

	Total	Administration & Management	Roads & Transport	Planning & Environment
	(1)	(2)	(3)	(4)
Grants	0.87***	0.38**	0.82***	0.72***
	(0.09)	(0.17)	(0.14)	(0.14)
Altitude	5.09***	1.17	2.09***	0.69
	(1.40)	(0.90)	(0.72)	(1.00)
Population	-0.07*	-0.06*	-0.03**	-0.03*
	(0.04)	(0.03)	(0.02)	(0.02)
Houses	2,620.32***	321.17	585.34	1,287.76**
	(910.86)	(535.36)	(379.05)	(500.74)
Aged	2,814.87	2,279.31	-2,118.45	7,334.39
	(9,310.60)	(6,303.39)	(4,681.13)	(5,380.26)
Children	8,782.41	-6,071.13	-8,850.30	10,936.83
	(30,648.31)	(18,043.98)	(20,418.58)	(16,040.36)
Population Density	-0.96	-0.30	0.33	-0.30
	(0.87)	(0.46)	(0.38)	(0.47)
Population growth	3,748.01	-2,319.11	-2,362.06	-411.85
	(4,317.05)	(3,293.26)	(2,347.50)	(3,111.54)
Total employees	4,471.62	1,862.30	1,032.61	103.64
	(3,204.03)	(2,092.94)	(1,218.65)	(1,670.92)
Local unit	16,134.72	6,497.92	-355.68	240.67
	(12,747.64)	(7,280.13)	(5,619.08)	(7,940.58)
Outliers dummy	9,698.34**	8,416.57***	7,551.73***	8,579.50***
	(4,124.96)	(1,247.66)	(1,049.15)	(1,661.39)
Metropolitan dummy	3,700.99	1,266.15	963.65	453.63
	(2,435.73)	(1,379.38)	(1,120.07)	(1,083.12)
Constant	-2,795.09	-479.78	1,198.57	-1,299.48
	(2,968.83)	(2,009.99)	(1,728.93)	(1,734.73)
Communities dummy	YES	YES	YES	YES
Observations	223	223	223	223
R-squared	0.91	0.72	0.86	0.82
Robust LM test for spatial error dependence	0.84	0.43	6.05**	5.44**
Robust LM test for spatial lag dependence	3.53*	0.60	9.50***	2.10

Notes: Robust standard errors are shown in parentheses. The spatial weights matrix used to compute the test is a binary, contiguity-based one, according to which two municipalities are neighbors if they share a common border and it is row-standardized. *** significant at 1%; ** significant at 5%; * significant at 10%.

Table A5: Non spatial model and interaction	1 with population –	OLS regression
---	---------------------	-----------------------

	Total	Administration & Management	Roads & Transport	Planning & Environment
	(1)	(2)	(3)	(4)
Neighbors spending * Population	-0.17	0.06	-0.48*	-0.26
	(0.15)	(0.20)	(0.27)	(0.17)
Grants	0.86***	0.38**	0.81***	0.71***
	(0.09)	(0.17)	(0.14)	(0.14)
Altitude	5.39***	1.12	2.40***	0.67
	(1.39)	(0.92)	(0.75)	(1.00)
Population	0.17	-0.09	0.13	0.04
	(0.22)	(0.09)	(0.09)	(0.05)
Houses	2,586.13***	324.51	520.00	1,282.95**
	(912.68)	(534.95)	(379.75)	(502.05)
Aged	2,950.51	2,244.79	-2,456.38	7,579.84
	(9,329.84)	(6,338.17)	(4,698.42)	(5,331.12)
Children	9,516.40	-6,226.33	-8,470.91	11,118.14
	(30,790.85)	(18,158.54)	(20,440.38)	(16,034.68)
Population Density	-1.08	-0.28	0.26	-0.35
	(0.89)	(0.46)	(0.40)	(0.48)
Population growth	3,624.18	-2,326.46	-2,505.92	-445.23
	(4,315.08)	(3,303.50)	(2,342.38)	(3,114.20)
Total employees	4,394.32	1,899.96	1,037.81	109.83
	(3,213.64)	(2,123.14)	(1,208.27)	(1,665.94)
Local unit	16,959.44	6,358.39	-81.32	632.79
	(12,973.07)	(7,326.33)	(5,673.08)	(7,972.03)
Outliers dummy	9,645.76**	8,424.25***	7,541.33***	8,521.32***
	(4,133.69)	(1,252.12)	(1,047.43)	(1,667.64)
Metropolitan dummy	2,142.87	1,522.80	-121.78	-165.00
	(3,052.42)	(1,590.26)	(1,264.99)	(916.63)
Constant	-2,895.81	-395.41	1,230.17	-1,159.33
	(2,996.47)	(2,053.22)	(1,737.68)	(1,753.69)
Communities dummy	YES	YES	YES	YES
Observations	223	223	223	223
R-squared	0.91	0.72	0.86	0.82

Notes: Robust standard errors are shown in parentheses. *** significant at 1%; ** significant at 5%; * significant at 10%.

2011

2011/1, Oppedisano, V; Turati, G.: "What are the causes of educational inequalities and of their evolution over time in Europe? Evidence from PISA"

2011/2, Dahlberg, M; Edmark, K; Lundqvist, H.: "Ethnic diversity and preferences for redistribution"

2011/3, Canova, L.; Vaglio, A.: "Why do educated mothers matter? A model of parental help"

2011/4, Delgado, F.J.; Lago-Peñas, S.; Mayor, M.: "On the determinants of local tax rates: new evidence from Spain"

2011/5, Piolatto, A.; Schuett, F.: "A model of music piracy with popularity-dependent copying costs"

2011/6, Duch, N.; García-Estévez, J.; Parellada, M.: "Universities and regional economic growth in Spanish regions"

2011/7, Duch, N.; García-Estévez, J.: "Do universities affect firms' location decisions? Evidence from Spain"

2011/8, Dahlberg, M.; Mörk, E.: "Is there an election cycle in public employment? Separating time effects from election year effects"

2011/9, Costas-Pérez, E.; Solé-Ollé, A.; Sorribas-Navarro, P.: "Corruption scandals, press reporting, and accountability. Evidence from Spanish mayors"

2011/10, Choi, A.; Calero, J.; Escardíbul, J.O.: "Hell to touch the sky? Private tutoring and academic achievement in Korea"

2011/11, Mira Godinho, M.; Cartaxo, R.: "University patenting, licensing and technology transfer: how organizational context and available resources determine performance"

2011/12, Duch-Brown, N.; García-Quevedo, J.; Montolio, D.: "The link between public support and private R&D effort: What is the optimal subsidy?"

2011/13, Breuillé, M.L.; Duran-Vigneron, P.; Samson, A.L.: "To assemble to resemble? A study of tax disparities among French municipalities"

2011/14, McCann, P.; Ortega-Argilés, R.: "Smart specialisation, regional growth and applications to EU cohesion policy"

2011/15, Montolio, D.; Trillas, F.: "Regulatory federalism and industrial policy in broadband telecommunications"

2011/16, Pelegrín, A.; Bolancé, C.: "Offshoring and company characteristics: some evidence from the analysis of Spanish firm data"

2011/17, Lin, C.: "Give me your wired and your highly skilled: measuring the impact of immigration policy on employers and shareholders"

2011/18, **Bianchini**, **L.**; **Revelli**, **F.**: "Green polities: urban environmental performance and government popularity" **2011/19**, **López Real**, **J.**: "Family reunification or point-based immigration system? The case of the U.S. and Mexico"

2011/20, Bogliacino, F.; Piva, M.; Vivarelli, M.: "The impact of R&D on employment in Europe: a firm-level analysis"

2011/21, Tonello, M.: "Mechanisms of peer interactions between native and non-native students: rejection or integration?"

2011/22, García-Quevedo, J.; Mas-Verdú, F.; Montolio, D.: "What type of innovative firms acquire knowledge intensive services and from which suppliers?"

2011/23, Banal-Estañol, A.; Macho-Stadler, I.; Pérez-Castrillo, D.: "Research output from university-industry collaborative projects"

2011/24, Ligthart, J.E.; Van Oudheusden, P.: "In government we trust: the role of fiscal decentralization"

2011/25, Mongrain, S.; Wilson, J.D.: "Tax competition with heterogeneous capital mobility"

2011/26, Caruso, R.; Costa, J.; Ricciuti, R.: "The probability of military rule in Africa, 1970-2007"

2011/27, Solé-Ollé, A.; Viladecans-Marsal, E.: "Local spending and the housing boom"

2011/28, Simón, H.; Ramos, R.; Sanromá, E.: "Occupational mobility of immigrants in a low skilled economy. The Spanish case"

2011/29, Piolatto, A.; Trotin, G.: "Optimal tax enforcement under prospect theory"

2011/30, Montolio, D; Piolatto, A.: "Financing public education when altruistic agents have retirement concerns"

2011/31, García-Quevedo, J.; Pellegrino, G.; Vivarelli, M.: "The determinants of YICs' R&D activity"

2011/32, Goodspeed, T.J.: "Corruption, accountability, and decentralization: theory and evidence from Mexico"

2011/33, **Pedraja**, **F.**; **Cordero**, **J.M.**: "Analysis of alternative proposals to reform the Spanish intergovernmental transfer system for municipalities"

2011/34, Jofre-Monseny, J.; Sorribas-Navarro, P.; Vázquez-Grenno, J.: "Welfare spending and ethnic heterogeneity: evidence from a massive immigration wave"

2011/35, Lyytikäinen, T.: "Tax competition among local governments: evidence from a property tax reform in Finland"

2011/36, Brülhart, M.; Schmidheiny, K.: "Estimating the Rivalness of State-Level Inward FDI"

2011/37, García-Pérez, J.I.; Hidalgo-Hidalgo, M.; Robles-Zurita, J.A.: "Does grade retention affect achievement? Some evidence from Pisa"

2011/38, Boffa, f.; Panzar. J.: "Bottleneck co-ownership as a regulatory alternative"

2011/39, González-Val, R.; Olmo, J.: "Growth in a cross-section of cities: location, increasing returns or random growth?"

2011/40, Anesi, V.; De Donder, P.: "Voting under the threat of secession: accommodation vs. repression"

2011/41, Di Pietro, G.; Mora, T.: "The effect of the l'Aquila earthquake on labour market outcomes"

2011/42, Brueckner, J.K.; Neumark, D.: "Beaches, sunshine, and public-sector pay: theory and evidence on amenities and rent extraction by government workers"

2011/43, Cortés, D.: "Decentralization of government and contracting with the private sector"

2011/44, Turati, G.; Montolio, D.; Piacenza, M.: "Fiscal decentralisation, private school funding, and students' achievements. A tale from two Roman catholic countries"

2012

2012/1, Montolio, D.; Trujillo, E.: "What drives investment in telecommunications? The role of regulation, firms' internationalization and market knowledge"

2012/2, Giesen, K.; Suedekum, J.: "The size distribution across all "cities": a unifying approach"

2012/3, Foremny, D.; Riedel, N.: "Business taxes and the electoral cycle"

2012/4, García-Estévez, J.; Duch-Brown, N.: "Student graduation: to what extent does university expenditure matter?"

2012/5, Durán-Cabré, J.M.; Esteller-Moré, A.; Salvadori, L.: "Empirical evidence on horizontal competition in tax enforcement"

2012/6, Pickering, A.C.; Rockey, J.: "Ideology and the growth of US state government"

2012/7, Vergolini, L.; Zanini, N.: "How does aid matter? The effect of financial aid on university enrolment decisions"

2012/8, Backus, P.: "Gibrat's law and legacy for non-profit organisations: a non-parametric analysis"

2012/9, Jofre-Monseny, J.; Marín-López, R.; Viladecans-Marsal, E.: "What underlies localization and urbanization economies? Evidence from the location of new firms"

2012/10, Mantovani, A.; Vandekerckhove, J.: "The strategic interplay between bundling and merging in complementary markets"

2012/11, Garcia-López, M.A.: "Urban spatial structure, suburbanization and transportation in Barcelona"

2012/12, Revelli, F.: "Business taxation and economic performance in hierarchical government structures"

2012/13, Arqué-Castells, P.; Mohnen, P.: "Sunk costs, extensive R&D subsidies and permanent inducement effects"

2012/14, Boffa, F.; Piolatto, A.; Ponzetto, G.: "Centralization and accountability: theory and evidence from the Clean Air Act"

2012/15, Cheshire, P.C.; Hilber, C.A.L.; Kaplanis, I.: "Land use regulation and productivity – land matters: evidence from a UK supermarket chain"

2012/16, Choi, A.; Calero, J.: "The contribution of the disabled to the attainment of the Europe 2020 strategy headline targets"

2012/17, Silva, J.I.; Vázquez-Grenno, J.: "The ins and outs of unemployment in a two-tier labor market"

2012/18, González-Val, R.; Lanaspa, L.; Sanz, F.: "New evidence on Gibrat's law for cities"

2012/19, Vázquez-Grenno, J.: "Job search methods in times of crisis: native and immigrant strategies in Spain"

2012/20, Lessmann, C.: "Regional inequality and decentralization – an empirical analysis"

2012/21, Nuevo-Chiquero, A.: "Trends in shotgun marriages: the pill, the will or the cost?"

2012/22, Piil Damm, A.: "Neighborhood quality and labor market outcomes: evidence from quasi-random neighborhood assignment of immigrants"

2012/23, Ploeckl, F.: "Space, settlements, towns: the influence of geography and market access on settlement distribution and urbanization"

2012/24, **Algan**, **Y**.; **Hémet**, **C**.; **Laitin**, **D**.: "Diversity and local public goods: a natural experiment with exogenous residential allocation"

2012/25, Martinez, D.; Sjögren, T.: "Vertical externalities with lump-sum taxes: how much difference does unemployment make?"

2012/26, Cubel, M.; Sanchez-Pages, S.: "The effect of within-group inequality in a conflict against a unitary threat"

2012/27, Andini, M.; De Blasio, G.; Duranton, G.; Strange, W.C.: "Marshallian labor market pooling: evidence from Italy"

2012/28, Solé-Ollé, A.; Viladecans-Marsal, E.: "Do political parties matter for local land use policies?"

2012/29, Buonanno, P.; Durante, R.; Prarolo, G.; Vanin, P.: "Poor institutions, rich mines: resource curse and the origins of the Sicilian mafia"

2012/30, Anghel, B.; Cabrales, A.; Carro, J.M.: "Evaluating a bilingual education program in Spain: the impact beyond foreign language learning"

2012/31, Curto-Grau, M.; Solé-Ollé, A.; Sorribas-Navarro, P.: "Partisan targeting of inter-governmental transfers & state interference in local elections: evidence from Spain"

2012/32, Kappeler, A.; Solé-Ollé, A.; Stephan, A.; Välilä, T.: "Does fiscal decentralization foster regional investment in productive infrastructure?"

2012/33, Rizzo, L.; Zanardi, A.: "Single vs double ballot and party coalitions: the impact on fiscal policy. Evidence from Italy"

2012/34, Ramachandran, R.: "Language use in education and primary schooling attainment: evidence from a natural experiment in Ethiopia"

2012/35, Rothstein, J.: "Teacher quality policy when supply matters"

2012/36, Ahlfeldt, G.M.: "The hidden dimensions of urbanity"

2012/37, Mora, T.; Gil, J.; Sicras-Mainar, A.: "The influence of BMI, obesity and overweight on medical costs: a panel data approach"

2012/38, Pelegrín, A.; García-Quevedo, J.: "Which firms are involved in foreign vertical integration?"

2012/39, Agasisti, T.; Longobardi, S.: "Inequality in education: can Italian disadvantaged students close the gap? A focus on resilience in the Italian school system"

2013

2013/1, Sánchez-Vidal, M.; González-Val, R.; Viladecans-Marsal, E.: "Sequential city growth in the US: does age matter?"

2013/2, Hortas Rico, M.: "Sprawl, blight and the role of urban containment policies. Evidence from US cities"

2013/3, Lampón, J.F.; Cabanelas-Lorenzo, P-; Lago-Peñas, S.: "Why firms relocate their production overseas? The answer lies inside: corporate, logistic and technological determinants"

2013/4, Montolio, D.; Planells, S.: "Does tourism boost criminal activity? Evidence from a top touristic country"

2013/5, Garcia-López, M.A.; Holl, A.; Viladecans-Marsal, E.: "Suburbanization and highways: when the Romans, the Bourbons and the first cars still shape Spanish cities"

2013/6, Bosch, N.; Espasa, M.; Montolio, D.: "Should large Spanish municipalities be financially compensated? Costs and benefits of being a capital/central municipality"

2013/7, Escardíbul, J.O.; Mora, T.: "Teacher gender and student performance in mathematics. Evidence from Catalonia"

2013/8, Arqué-Castells, P.; Viladecans-Marsal, E.: "Banking towards development: evidence from the Spanish banking expansion plan"

2013/9, Asensio, J.; Gómez-Lobo, A.; Matas, A.: "How effective are policies to reduce gasoline consumption? Evaluating a quasi-natural experiment in Spain"

2013/10, Jofre-Monseny, J.: "The effects of unemployment benefits on migration in lagging regions"

2013/11, Segarra, A.; García-Quevedo, J.; Teruel, M.: "Financial constraints and the failure of innovation projects"

2013/12, Jerrim, J.; Choi, A.: "The mathematics skills of school children: How does England compare to the high performing East Asian jurisdictions?"

2013/13, González-Val, R.; Tirado-Fabregat, D.A.; Viladecans-Marsal, E.: "Market potential and city growth: Spain 1860-1960"

2013/14, Lundqvist, H.: "Is it worth it? On the returns to holding political office"

2013/15, Ahlfeldt, G.M.; Maennig, W.: "Homevoters vs. leasevoters: a spatial analysis of airport effects"

2013/16, Lampón, J.F.; Lago-Peñas, S.: "Factors behind international relocation and changes in production

geography in the European automobile components industry"

2013/17, **Guío**, **J.M.**; **Choi**, **A.**: "Evolution of the school failure risk during the 2000 decade in Spain: analysis of Pisa results with a two-level logistic mode"

2013/18, Dahlby, B.; Rodden, J.: "A political economy model of the vertical fiscal gap and vertical fiscal imbalances in a federation"

2013/19, Acacia, F.; Cubel, M.: "Strategic voting and happiness"

2013/20, Hellerstein, J.K.; Kutzbach, M.J.; Neumark, D.: "Do labor market networks have an important spatial dimension?"

2013/21, Pellegrino, G.; Savona, M.: "Is money all? Financing versus knowledge and demand constraints to innovation"

2013/22, Lin, J.: "Regional resilience"

2013/23, Costa-Campi, M.T.; Duch-Brown, N.; García-Quevedo, J.: "R&D drivers and obstacles to innovation in the energy industry"

2013/24, Huisman, R.; Stradnic, V.; Westgaard, S.: "Renewable energy and electricity prices: indirect empirical evidence from hydro power"

2013/25, Dargaud, E.; Mantovani, A.; Reggiani, C.: "The fight against cartels: a transatlantic perspective"

2013/26, Lambertini, L.; Mantovani, A.: "Feedback equilibria in a dynamic renewable resource oligopoly: preemption, voracity and exhaustion" 2013/27, Feld, L.P.; Kalb, A.; Moessinger, M.D.; Osterloh, S.: "Sovereign bond market reactions to fiscal rules and no-bailout clauses – the Swiss experience"

2013/28, Hilber, C.A.L.; Vermeulen, W.: "The impact of supply constraints on house prices in England"

2013/29, Revelli, F .: "Tax limits and local democracy"

2013/30, Wang, R.; Wang, W.: "Dress-up contest: a dark side of fiscal decentralization"

2013/31, Dargaud, E.; Mantovani, A.; Reggiani, C.: "The fight against cartels: a transatlantic perspective"

2013/32, Saarimaa, T.; Tukiainen, J.: "Local representation and strategic voting: evidence from electoral boundary reforms"

2013/33, Agasisti, T.; Murtinu, S.: "Are we wasting public money? No! The effects of grants on Italian university students' performances"

2013/34, Flacher, D.; Harari-Kermadec, H.; Moulin, L.: "Financing higher education: a contributory scheme"

2013/35, Carozzi, F.; Repetto, L.: "Sending the pork home: birth town bias in transfers to Italian municipalities"

2013/36, Coad, A.; Frankish, J.S.; Roberts, R.G.; Storey, D.J.: "New venture survival and growth: Does the fog lift?"

2013/37, **Giulietti**, **M.**; **Grossi**, **L.**; **Waterson**, **M.**: "Revenues from storage in a competitive electricity market: Empirical evidence from Great Britain"

2014

2014/1, Montolio, D.; Planells-Struse, S.: "When police patrols matter. The effect of police proximity on citizens' crime risk perception"

2014/2, Garcia-López, M.A.; Solé-Ollé, A.; Viladecans-Marsal, E.: "Do land use policies follow road construction?"

2014/3, Piolatto, A.; Rablen, M.D.: "Prospect theory and tax evasion: a reconsideration of the Yitzhaki puzzle"

2014/4, Cuberes, D.; González-Val, R.: "The effect of the Spanish Reconquest on Iberian Cities"

2014/5, Durán-Cabré, J.M.; Esteller-Moré, E.: "Tax professionals' view of the Spanish tax system: efficiency, equity and tax planning"

2014/6, Cubel, M.; Sanchez-Pages, S.: "Difference-form group contests"

2014/7, Del Rey, E.; Racionero, M.: "Choosing the type of income-contingent loan: risk-sharing versus risk-pooling"

2014/8, Torregrosa Hetland, S.: "A fiscal revolution? Progressivity in the Spanish tax system, 1960-1990"

2014/9, Piolatto, A.: "Itemised deductions: a device to reduce tax evasion"

2014/10, Costa, M.T.; García-Quevedo, J.; Segarra, A.: "Energy efficiency determinants: an empirical analysis of Spanish innovative firms"

2014/11, García-Quevedo, J.; Pellegrino, G.; Savona, M.: "Reviving demand-pull perspectives: the effect of demand uncertainty and stagnancy on R&D strategy"

2014/12, Calero, J.; Escardíbul, J.O.: "Barriers to non-formal professional training in Spain in periods of economic growth and crisis. An analysis with special attention to the effect of the previous human capital of workers"

2014/13, Cubel, M.; Sanchez-Pages, S.: "Gender differences and stereotypes in the beauty"

2014/14, Piolatto, A.; Schuett, F.: "Media competition and electoral politics"

2014/15, Montolio, D.; Trillas, F.; Trujillo-Baute, E.: "Regulatory environment and firm performance in EU telecommunications services"

2014/16, **Lopez-Rodriguez**, **J.**; **Martinez**, **D.**: "Beyond the R&D effects on innovation: the contribution of non-R&D activities to TFP growth in the EU"

2014/17, González-Val, R.: "Cross-sectional growth in US cities from 1990 to 2000"

2014/18, Vona, F.; Nicolli, F.: "Energy market liberalization and renewable energy policies in OECD countries"

2014/19, Curto-Grau, M.: "Voters' responsiveness to public employment policies"

2014/20, Duro, J.A.; Teixidó-Figueras, J.; Padilla, E.: "The causal factors of international inequality in co2 emissions per capita: a regression-based inequality decomposition analysis"

2014/21, Fleten, S.E.; Huisman, R.; Kilic, M.; Pennings, E.; Westgaard, S.: "Electricity futures prices: time varying sensitivity to fundamentals"

2014/22, Afcha, S.; García-Quevedo, J.: "The impact of R&D subsidies on R&D employment composition"

2014/23, Mir-Artigues, P.; del Río, P.: "Combining tariffs, investment subsidies and soft loans in a renewable electricity deployment policy"

2014/24, Romero-Jordán, D.; del Río, P.; Peñasco, C.: "Household electricity demand in Spanish regions. Public policy implications"

2014/25, Salinas, P.: "The effect of decentralization on educational outcomes: real autonomy matters!"

2014/26, Solé-Ollé, A.; Sorribas-Navarro, P.: "Does corruption erode trust in government? Evidence from a recent surge of local scandals in Spain"

2014/27, Costas-Pérez, E.: "Political corruption and voter turnout: mobilization or disaffection?"

2014/28, Cubel, M.; Nuevo-Chiquero, A.; Sanchez-Pages, S.; Vidal-Fernandez, M.: "Do personality traits affect productivity? Evidence from the LAB"

2014/29, Teresa Costa, M.T.; Trujillo-Baute, E.: "Retail price effects of feed-in tariff regulation"

2014/30, Kilic, M.; Trujillo-Baute, E.: "The stabilizing effect of hydro reservoir levels on intraday power prices under wind forecast errors"

2014/31, Costa-Campi, M.T.; Duch-Brown, N.: "The diffusion of patented oil and gas technology with environmental uses: a forward patent citation analysis"

2014/32, Ramos, R.; Sanromá, E.; Simón, H.: "Public-private sector wage differentials by type of contract: evidence from Spain"

2014/33, Backus, P.; Esteller-Moré, A.: "Is income redistribution a form of insurance, a public good or both?"

2014/34, Huisman, R.; Trujillo-Baute, E.: "Costs of power supply flexibility: the indirect impact of a Spanish policy change"

2014/35, Jerrim, J.; Choi, A.; Simancas Rodríguez, R.: "Two-sample two-stage least squares (TSTSLS) estimates of earnings mobility: how consistent are they?"

2014/36, Mantovani, A.; Tarola, O.; Vergari, C.: "Hedonic quality, social norms, and environmental campaigns"



Des campus d'envel·lèncie in BIKC 🚟 🔤 🛄



ieb@ub.edu www.ieb.ub.edu

Fiscal Federalism